Author's response to reviews

Title: Nurse practitioner interactions in acute and long-term care: An exploration of the role of knotworking in supporting interprofessional collaboration

Authors:

Christina Hurlock-Chorostecki (Tina.HurlockChorostecki@lhsc.on.ca)
Mary van Soeren (mary.van.soeren@dal.ca)
Kathleen MacMillan (kathleen.macmillan@dal.ca)
Faith Donald (fdonald@ryerson.ca)
Souraya Sidani (ssidani@ryerson.ca)
Scott Reeves (s.reeves@sgul.kingston.ac.uk)

Version: 5 Date: 25 September 2015

Author's response to reviews: see over
Dear Ms. Marielette Costoy;

Thank you for accepting our research article, Nurse practitioner interactions in acute and long-term care: An exploration of the role of knotworking in supporting interprofessional collaboration, for publication in BMC Nursing.

The following requests provided by you have been addressed:

1. Email addresses of each author now appear on the title page. The title page has been adjusted to the style indicated online.
2. A section titled “List of Abbreviations” was added to the manuscript after the discussion. The abbreviations used in the article are listed.
3. Informed written consent has been inserted into the paragraph on Setting and Sample.
4. This final version of the article has been read and approved by the authors. This is noted in a sentence within the Author Contributions.
5. The word Objective has been removed from the abstract and replaced with “Background”.
6. Section headings were changed as requested to be “Acknowledgements” and “Competing Interests”.
7. A section entitled Figure Legend was added after the References. Here the figure number, title of figure and a less than 300 word legend was inserted.
8. The example online for journal style was reviewed and the document updated. The reference list was reviewed and updated. End Note was used to ensure compliance with the BMC Nursing referencing style.

Sincerely,

Scott Reeves